Author’s response to reviews

Title: Effectiveness of Primary Care Triple P on Child Psychosocial Problems in Preventive Child Healthcare: a Randomized Controlled Trial

Authors:

Willem Spijkers (w.spijkers@med.umcg.nl)
Daniëlle EMC Jansen (d.e.m.c.jansen@umcg.nl)
Sijmen A Reijneveld (s.a.reijneveld@umcg.nl)

Version: 4 Date: 31 August 2013

Author's response to reviews: see over
Dear Editor, dear Dr. Salam,

Attached please find our revised manuscript, "Effectiveness of Primary Care Triple P on Child Psychosocial Problems in Preventive Child Healthcare: a Randomized Controlled Trial", to be considered for publication as an article in BMC Medicine. We appreciate the opportunity to revise our manuscript and found the comments of the reviewers very helpful in improving the clarity of our manuscript.

We are confident that we have successfully addressed all the items the reviewers recommended us to revise. We have done this by describing our responses, preceded by [RESPONSE], and followed by an indication of the pages of the text that have been modified in response to each of these comments. In our manuscript, we highlighted the changes by using the track changes mode in MS Word. We hope that this revised version can contribute to the contents of BMC Medicine. We look forward to hearing from you.

With kind regards,
also on behalf of the other authors (D.E.M.C. Jansen, PhD, and Prof. S.A. Reijneveld, MD PhD),

W. Spijkers, MSc.

Address for correspondence:
W. Spijkers, MSc.
University Medical Center Groningen, University of Groningen, Department of Health Sciences, Groningen, the Netherlands
PO Box 196
9700 AD Groningen
The Netherlands
w.spijkers@umcg.nl
tel.+31 (0)50 3638369
fax +31 (0)50 3636251